Author’s response to reviews

Title: Safety checklists for use by medical care teams in acute hospital settings

Authors:

Henry CH Ko (henry.ko@ctc.usyd.edu.au)
Tari J Turner (tari.turner@monash.edu)
Monica A Finnigan (monica.finnigan@southernhealth.org.au)

Version: 6 Date: 22 August 2011

Author’s response to reviews:

Dear BMC HSR,

This is to confirm that all authors have read and approved the final manuscript.

Kind Regards,

Henry Ko
(first / corresponding author)
Date: 22/8/2011